Author’s response to reviews

Title: Machine Learning Classification of First-Episode Schizophrenia Spectrum Disorders and Controls using Whole Brain White Matter Fractional Anisotropy

Authors:
Pavol Mikolas (pavol.mikolas@gmail.com)
Jaroslav Hlinka (jaroslav.hlinka@nudz.cz)
Antonin Skoch (ansk@medicon.cz)
Pitra Zbynek (z.pitra@gmail.com)
Thomas Frodl (thomas.frodl@med.ovgu.de)
Filip Spaniel (filip.spaniel@nudz.cz)
Tomas Hajek (tomas.hajek@dal.ca)

Version: 5 Date: 20 Mar 2018

Author’s response to reviews:

Dear Dr. Hickey,

thank you for your essential corrections, we rewrote the suggested passages.

Additionally we would like to explain, that the section Materials and Methods - 2.1 Subjects is very essential to the understanding of the definition of FES in our analyses and was also subjected to the reviewers comments. Therefore we include the full explanation of the inclusion criteria rather than a citation of our previous work.

Thank you for your understanding.
Best regards

Mikolas